Author's response to reviews

Title: A taboo within a stigma? A qualitative study of managing incontinence in people with dementia living at home

Authors:

Vari M Drennan (v.drennan@sgul.kingston.ac.uk)
Laura Cole (laura.cole@sgul.kingston.ac.uk)
Steve Iliffe (s.iliffe@ucl.ac.uk)

Version: 3 Date: 11 November 2011

Author's response to reviews: see over
Dear editor,

Thank you for deciding to publish this paper. We have made the following requested formatting changes. Our changes are detailed in bold against the request below.

Yours sincerely

Vari Drennan

---

**Formatting changes to be made:**

**Major revisions** (we require the author to make these changes)

**Ethical approval:** Please provide the name of the NHS ethics committee that gave approval for your research (please do not simply refer to them as 'local'). *This has been amended*

**Tables:** we are unable to display shading and colours within the tables. Can we therefore ask you to use another method of highlighting the shaded text e.g. by making it bold. *There is no colour or shading in this table.*

Please place all tables in a separate section at the end of the manuscript, following the reference list. The table title should appear above the table and should be only one sentence long. Please place any other text as a legend (or key) below the table. *The table has been moved to the end of the manuscript with the table title.*

**Authors' contributions:** Please indicate that all authors read and approved the final manuscript. *This has been added*

**References:** please provide a functioning link to the website cited in the acknowledgements (http://www.evidem.org.uk) *This has been corrected to http://www.evidem.org.uk*

**Minor revisions** (we can make these changes for you, although it will speed up publication of your manuscript if you do them while making the major changes above)

**Highlighting/tracking:** Please remove all highlighting and tracking from the manuscript, and change any coloured text to black. *This has been changed to black*

Please remove footer (*"A taboo within a stigma?"*) *This has been removed.*